Author’s response to reviews

Title: Dental caries prevalence, knowledge, attitude and preventive practice on oral health among school children of Chepang communities in Nepal.

Authors:

Lonim Prasai Dixit Dr (lprasai@yahoo.com)
Ajay Shakya Dr (drajayshakya@gmail.com)
Manash Shrestha Dr (manashshrestha@yahoo.com)
Ayush Shrestha Dr (hsuyayush@gmail.com)

Version: 4 Date: 28 September 2012

Author's response to reviews: see over
Author's covering letter for initial submission

Title: Dental caries prevalence, knowledge, attitude and preventive practice on oral health among school children of Chepang communities in Nepal.

Authors:

Version: 1 Date: 4 September 2012

Comments:

Dear members of the editorial team,

Thank you very much for the opportunity to present to you the study on Chepang Communities. Chepangs are one of the most disadvantaged minority groups living in remote hills. The total population is approximately 55,000 plus.

The study is intended as an advocacy tool to strengthen and integrate oral health component into school health programs. Each school needed about 1-4 hours walk uphill where oral health services such as extraction, ART was also conducted.

This would also be an opportunity for public health personnel like us from developing countries to present on our oral health situation although the study is limited to a certain community.

I would like to request you to kindly consider this article. I am very open to any changes and advise.

Regards,

Lonim.